Author’s response to reviews

Title: Periodic health visits by primary care practice model, a population-based study using health administrative data.

Authors:
Natasha Saunders (natasha.saunders@sickkids.ca)
Jun Guan (jun.guan@ices.on.ca)
Longdi Fu (longdi.fu@ices.on.ca)
Helen Guo (helen.guo@ices.on.ca)
Xuesong Wang (xuesong@ices.on.ca)
Astrid Guttmann (astrid.guttmann@ices.on.ca)

Version: 3 Date: 10 Jan 2019

Author’s response to reviews:

Dr. Natasha Saunders
The Hospital for Sick Children
555 University Avenue
Toronto, Ontario M5G 1X8

January 10th, 2019

Dr. Tovah Honor Aronin, BMC Family Practice Editor

Dear Dr. Aronin,

Please find enclosed a revised manuscript entitled; “Periodic health visits by primary care practice model, a population-based study using health administrative data.” (FAMP-D-17-
00391R3) which we are re-submitting for exclusive consideration of publication as a Research Article in the BMC Family Practice.

We thank the editors and reviewers for their consideration of our manuscript. We have added the tables as requested to the second revision of this manuscript.

We look forward to hearing from you regarding our submission. We would be glad to respond to any further questions and comments that you may have. Please address all correspondence concerning this manuscript to me at The Hospital for Sick Children and feel free to correspond with me by email at natasha.saunders@sickkids.ca.

Sincerely,

Natasha Saunders

Staff Pediatrician and Associate Scientist, The Hospital for Sick Children
Adjunct Scientist, The Institute for Clinical Evaluative Science